Author's response to reviews

Title: Post-partum, post-sterilisation tubo-ovarian abscess caused by Fusobacterium necrophorum: a case report

Authors:

Chenchit Chayachinda (bananachenge@yahoo.com)
Amornrut Leelaporn (siale@mahidol.ac.th)
Pornpimol Ruangvutilert (siprv@mahidol.ac.th)
Manopchai Thamkhantho (simth@mahidol.ac.th)

Version: 10 Date: 27 August 2012

Author's response to reviews: see over
Authors’ response to reviews

Title: Post-partum, post-sterilisation tubo-ovarian abscess caused by Fusobacterium necrophorum subsp. funduliforme: a case report

Authors: Chenchit Chayachinda MD1*
         Amornrut Leelaporn MD2
         Pornpimol Ruangvutilert MD1
         Manopchai Thamkhantho MD1

Version: 9
Date: August 27th, 2012
Authors’ response to reviewers: see over
Dear The Biomed Central Editorial Team

Object: MS ID 6758672127081538 Post-partum, post-sterilisation tubo-ovarian abscess caused by *Fusobacterium necrophorum subsp. Funduliforme*: a case report

Thank you for consideration of our manuscript for publication in your journal. We have reviewed the above manuscript according to your reviewers’ comments.

Comments from the Editorial team

1. Please replace your title with: Post-partum, post-sterilisation tubo-ovarian abscess caused by *Fusobacterium necrophorum*: a case report

Reply: The authors replace the previous title to the one suggested by the editorial team.

______________________________________________________________